Author’s response to reviews

Title: Utilizing real-time contrast medium to detect the fistula of giant spinal arachnoid cyst and treat with minimal invasive surgery

Authors:

Guang-Yu Ying (ygy@zju.edu.cn)
Kai-Sheng Chang (180567@cch.org.tw)
Ya-Juan Tang (anthaena.tang@hotmail.com)
Chun-Yuan Cheng (83998@cch.org.tw)
Yong-Jian Zhu (neurosurgery@zju.edu.cn)
Chien-Min Chen (cmchen.cch@gmail.com)

Version: 2 Date: 04 Jan 2019

Author’s response to reviews:

4 January 20189

Dear Editor,

Title: Utilizing Real-Time Contrast Medium to Detect the Fistula of Giant Spinal Arachnoid Cyst and Treat with Minimal Invasive Surgery

Manuscript ID: BSUR-D-18-00337

Authors: Guang-Yu Ying; Kai-Sheng Chang; Ya-Juan Tang; Chun-Yuan Cheng; Yong -Jian Zhu; Chien-Min Chen

As Editorial requests, we have made the following point-by-point changes to our manuscript. We hope that the Editorial Staff will find this revised manuscript suitable for publication in the BMC Surgery. We thank you for your time and consideration.

1. According to our submission guidelines, we require that Case reports submitted to BMC have the following sections: Background, Case presentation, and Discussion and Conclusions. Therefore, please combine your 'Discussion' and 'Conclusion' sections into one section.

Answer: Discussion and Conclusion sections have been combined into one section.
2. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case

Answer: We have added following statements in the Authors' contributions section.

Authors' contributions

KSC, GYY, YJZ, CYC and CMC contributed to the study conception. KSC, GYY and CMC drafted the manuscript. YJT contributed to the study design, analysis and interpretation of data. YJZ, CYC and CMC revised the manuscript critically for important intellectual content. All authors read and approved the final manuscript.

We hope that you will find this revised manuscript suitable for publication in the BMC Surgery. We thank you for your time and consideration.

Sincerely yours,

Chien-Min Chen

Division of Neurosurgery
Dept. of Surgery
Changhua Christian Hospital
No.135, Nansiao St. Changhua city, Changhua county, Taiwan
Fax: +886-04-7228289